Author’s response to reviews

Title: Neuromuscular shoulder activity during exercises with different combinations of stable and unstable weight mass

Authors:

Omar Baritello (baritello.o@gmail.com; baritello@uni-potsdam.de)
Mina Khajooei (khajooei@uni-potsdam.de)
Tilman Engel (tiengel@uni-potsdam.de)
Stephan Kopinski (kopinski@uni-potsdam.de)
Andrew Quarmby (andrewquarmby17@gmail.com)
Steffen Mueller (stef.mueller@hochschule-trier.de)
Frank Mayer (fmayer@uni-potsdam.de)

Version: 6 Date: 02 Mar 2020

Author’s response to reviews:

Dear Editor

referring to the Authorship of AQ:

The author AQ made substantial contribution reviewing the Manuscript (Grammar check and spelling) of the final version. Moreover he took part in 2 measurements (out of 16) and he help in some data calculations. However his main contribution was on the Proof of English.

Would you please therefore accept the new description of his contribution "manuscript revision and data interpretation". I really would like to keep him inside the authorship list.

Thank you so much for your support and comprehension.

Best greetings

Omar Baritello